Title: Incidence Rates of Sickness Absence Related to Mental Disorders: A Systematic Literature Review

Authors:

Carolyn S Dewa (carolyn_dewa@camh.net)
Desmond Loong (desmond.loong@camh.ca)
Sarah Bonato (sarah.bonato@camh.ca)
Hiske Hees (h.l.hees@amc.uva.nl)

Version: 4
Date: 12 February 2014

Author's response to reviews: see over
February 11, 2014

Dear Dr. Pafitis:

We received your email informing us that, “Further consideration of your manuscript is conditional on improvement of the English used.” I was somewhat concerned by your request given that I am a native English speaker and one of my degrees in English Literature. When I asked colleagues who also are native English speakers to read the manuscript, they did not find the “style of the written English” to be a hindrance. In light of the fact that your request was not accompanied with specific suggestions, I have done my best to simplify the language used in the manuscript. If there is something in particular that you feel should be revised, please identify the text and I will be happy to revise it.

I can be reached at:

Carolyn S. Dewa, MPH, PhD.
Head, Centre for Research on Employment and Workplace Health
Centre for Addiction and Mental Health
33 Russell St., Room 1000
Toronto, Ontario M5S 2S1 Canada
(416) 535-8501 x30733
e-mail: carolyn.dewa@camh.ca

Thank you for your consideration.

Sincerely,

Carolyn S. Dewa, MPH, PhD
Head, Centre for Research on Employment and Workplace Health
Centre for Addiction and Mental Health

Professor, Department of Psychiatry, University of Toronto